Author's response to reviews

Title: Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Alison Lott (alott@ohri.ca)
Dean A Fergusson (daferfusson@ohri.ca)
Kaveh G Shojania (kshojania@ohri.ca)
Jonathan Kimmelman (jonathan.kimmelman@mcgill.ca)
Raphael Saginur (rsaginur@ottawahospital.on.ca)

Version: 3 Date: 23 June 2008

Author's response to reviews:

Dear Dr. Foy:

Re: 1509385366196781 – Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors: Jamie C. Brehaut, Alison Lott, Dean A. Fergusson, Kaveh G. Shojania, Jonathan Kimmelman, Raphael Saginur

Thank you for your comments. We have added in a reference on page 21 for R statistical software as you suggested. We accepted all of the copy-editing comments, and fixed references # 26, 54-6, 78, 84, 92-3, and 100. Word count of the main body of the text is now 5916.

Please let us know if you have any other questions or concerns.

Sincerely,

Jamie C. Brehaut, PhD